Author’s response to reviews

**Title:** Quantifying cognition at the bedside: A novel approach combining cognitive symptoms and signs in HIV

**Authors:**
Marie-Josée Brouillette (pozbhn.med@mcgill.ca)
Lesley Fellows (lesley.fellows@mcgill.ca)
Lisa Palladini (lisa.palladini@mail.mcgill.ca)
Lois Finch (lois.finch@mail.mcgill.ca)
Réjean Thomas (Rejean.Thomas@lactuel.ca)
Nancy Mayo (nancy.mayo@mcgill.ca)

**Version:** 3  **Date:** 14 Oct 2015

**Author’s response to reviews:**

October 13, 2015

Iskra Staneva,
Assistant Editor at BioMed Central

NURL-D-15-00074R2
Quantifying cognition at the bedside: A novel approach combining cognitive symptoms and signs in HIV

Marie-Josée Brouillette, M.D.; Lesley K. Fellows, MD, DPhil; Lisa Palladini, MSc; Lois Finch, PhD; Réjean Thomas, MD; Nancy E. Mayo, PhD
BMC Neurology

Dear Dr. Staneva,

Please find attached our manuscript entitled “Quantifying cognition at the bedside: A novel approach combining cognitive symptoms and signs in HIV” (NURL-D-15-00074R2), currently under review for publication in BMC Neurology.

The editorial comments have been addressed and a revised version of the manuscript has been submitted. The changes are listed below:
1. Email addresses: The email addresses of all authors appear on the title page.

2. Funding information: The funding information has been moved to the Acknowledgements section.

3. Tables: The tables have been moved to the end of the manuscript text file.

4. Tracked changes: The version with the tracked changes has been removed. All changes to the revised manuscript are indicated in the text by highlighting.

We hope that these changes are to your satisfaction.

Respectfully submitted,

Marie-Josée Brouillette, MD, FRCRC
Associate Professor
McGill University
Montreal, Quebec, Canada